Reviewer #1:

**Scientific Quality:** Grade C (Good)

**Language Quality:** Grade B (Minor language polishing)

**Conclusion:** Minor revision

**Specific Comments to Authors:** Postmenopausal bleeding and thickened endometrium in menopausal women indicate the presence of endometrial lesions. In this study, a total of 187 postmenopausal women with bleeding or asymptomatic thickened endometrium underwent diagnostic hysteroscopy. The authors found that diagnostic hysteroscopy is recommended based on its lower cost, lower rate of complications, and high accuracy. The study is well designed and the results are interesting. Minor comments: 1. The manuscript should be edited. Some minor language polishing should be revised. 2. The results are well discussed, however, the authors should also discuss the limit of the study.

Answer: 1. The manuscript has been language edited by MedE Medical Editing Group. 2. The limit of the study has been discussed.

---

(2) **Company editor-in-chief:**

I have reviewed the Peer-Review Report, full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

Answer: The funding agency copy has been added.